Author's response to reviews

Title: The Construct Validity and Reliability of the Turkish Version of Spreitzer's Psychological Empowerment Scale

Authors:

Sarp Uner (sarpuner@hacettepe.edu.tr)
Sevgi Turan (sturan@hacettepe.edu.tr)

Version: 4 Date: 11 February 2010

Author's response to reviews:

Dear editor
Many thanks for your kindly suggestions
In the study, we explain the aim of the study in the cover letter of the questionnaire and subjects voluntarily participated to the present study. According to your recommendations we add some statements about informed consent of participants to the methods/procedure section of the manuscript.
Best wishes
Dr. Sarp Uner